Author’s response to reviews

Title: Severe but reversible pulmonary hypertension in scleromyxedema and multiple myeloma

Authors:

Mazen Kreidy (mazenkreidy@hotmail.com)
Ali Al-Hilli (al-hilli.ali@marshfieldclinic.org)
Ralph Yachoui (Ralph_yachoui@hotmail.com)
Jeffrey Resnick (resnick.jeffrey@marshfieldclinic.org)

Version: 5 Date: 04 Dec 2019

Author’s response to reviews:

Dear Editor

The prior Consent for publication was filled by our scientific writing department, who was under the impression the written consent was from the patient. We had a verbal consent from the patient before him passing away, and a written consent from the wife, obtained after the patient passing away. This has been corrected in the consent for Publication.

Sincerely

MK